Author’s response to reviews

Title: TLR4 Modulates Inflammatory Gene Targets in the Retina During Bacillus cereus Endophthalmitis

Authors:
Phillip Coburn (phillip-coburn@ouhsc.edu)
Frederick Miller (frederick-miller@ouhsc.edu)
Austin LaGrow (austin-lagrow@ouhsc.edu)
Salai Parkunan (madhu.reju27@gmail.com)
C. Randall (christopher-randall@ouhsc.edu)
Rachel Staats (rachel-staats@ouhsc.edu)
Michelle Callegan (michelle-callegan@ouhsc.edu)

Version: 4 Date: 27 Mar 2018

Author’s response to reviews:

1. Unfortunately, a citation for 'S2 Table' could not be found within the main text of your manuscript. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

See page 8, line 13 and page 11, line 12.

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Completed.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Completed.